Author's response to reviews

Title: A Statistical Framework for Quantifying Clinical Equipoise for Individual Cases during Randomized Controlled Surgical Trials

Authors:

Nicholas R Parsons (nick.parsons@warwick.ac.uk)
Yuri Kulikov (y.i.kulikov@warwick.ac.uk)
Alan Girling (a.j.girling@bham.ac.uk)
Damian Griffin (damian.griffin@warwick.ac.uk)

Version: 4 Date: 13 December 2011

Author's response to reviews:

Tracking removed and minor typographical errors corrected. Manuscript sections reformatted and repositioned to provide correct structure for a methodology paper.